Author's response to reviews

Title: Quality of life and metabolic status in mildly depressed patients with type 2 diabetes treated with paroxetine: A double-blind randomised placebo controlled 6-month trial

Authors:

Maria Paile-Hyvarinen (maria.paile@helsinki.fi)
Kristian Wahlbeck (kristian.wahlbeck@stakes.fi)
Johan G Eriksson (johan.eriksson@helsinki.fi)

Version: 6 Date: 8 June 2007

Author's response to reviews: